Author's response to reviews

Title: Principles of brain plasticity in improving sensorimotor function of the knee and leg in healthy subjects: A double-blind randomized exploratory trial

Authors:

Eva Ageberg (eva.ageberg@med.lu.se)
Anders Björkman (Anders.Bjorkman@med.lu.se)
Birgitta Rosén (Birgitta.Rosen@med.lu.se)
Göran Lundborg (Goran.Lundborg@med.lu.se)
Ewa M Roos (eroos@health.sdu.dk)

Version: 3 Date: 1 August 2009

Author's response to reviews:

Dear Editorial Team,

Thank you for your response. We hereby submit the revised version of the paper with the required formatting changes (please see point-by-point description below). We have made no changes in the manuscript in response to the comments made by Referee 2.

Sincerely,
Eva Ageberg

Formatting changes requested
--------------------------------------

Major revisions (We require the author to make these changes)

Reference 24, In press - Has this article now been published? If so, please provide the citation details.

AUTHORS’ RESPONSE AND ACTION
The article details are updated.

Table size - These must fit into an A4 portrait page. If you cannot re-orientate the table onto a single a4 portrait page (maximum size 176mm x 230mm), please move it to the additional files.

Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

AUTHORS’ RESPONSE AND ACTION
The Table is re-oriented onto a single A4 portrait page.

References please cite in numerical order within the main text. Reference 47 is currently only cited in the tables section.

AUTHORS’ RESPONSE AND ACTION
All references are now cited in the main text.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title - please remove boxes from around authors' email addresses and keywords from this section.

AUTHORS’ RESPONSE AND ACTION
These changes have been made.

Subjects and methods - please rename Methods.

AUTHORS’ RESPONSE AND ACTION
Changed.

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

AUTHORS’ RESPONSE AND ACTION
Changed.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

AUTHORS’ RESPONSE AND ACTION
The Figure is cropped and there is now no white space around the image.